Author's response to reviews

Title: Epidemiology of Musculoskeletal Upper Extremity Ambulatory Surgery in the United States

Authors:

Nitin B Jain (njain1@partners.org)
Laurence D Higgins (ldhiggins@partners.org)
Elena Losina (elosina@partners.org)
Jamie Collins (jcollins@partners.org)
Philip E Blazar (pblazar@partners.org)
Jeffrey N Katz (jnkatz@partners.org)

Version: 2 Date: 25 June 2013

Author's response to reviews: see over
Dear Mr. Aldea:

We have revised the manuscript as recommended. Specifically, we made the following changes:

1.) Please include a "Competing interests" Section:
This has been included in the revised manuscript (page 14).

2.) Please provide an "Authors' contributions" Section:
This is now included as per your guidelines (page 14)

3.) Please add "Acknowledgements" Section:
This has been added (page 14).

4.) Ethics statement:
This has been now included in the methods section (page 7).

Thank you considering our manuscript for publication in BMC Musculoskeletal Disorders.

Sincerely,

Nitin Jain, MD